Author's response to reviews

Title: Apolipoprotein L1, Income and Early Kidney Damage

Authors:

Ruth Tamrat (rtamrat1@jhmi.edu)
Carmen A. Peralta (carmenalicia.peralta@ucsf.edu)
Salman M. Tajuddin (salman.tajuddin@nih.gov)
Michele K. Evans (evansm@grc.nia.nih.gov)
Alan B. Zonderman (zondermana@mail.nih.gov)
Deidra C. Crews (dcrews1@jhmi.edu)

Version: 2
Date: 20 December 2014

Author's response to reviews:

Please see the attached